\begin{table}%t1
\caption{\label{dev}Development of the YOHKOH HXT Flare Catalogue.}
%\centerline
{
\small\begin{tabular}{ccccccccc}
 \hline\hline
 Version & & Latest & & \multicolumn{5}{c}{Contents} \\
 \cline{5-9}
  & References & event & Records & HXT & HXT & \textit{GOES} & SXT & WBS \\
  & & & & (lc)$^{a}$ & (i)$^{a}$ & & & \\
  \hline
  1 & \citet{kos93} & 1992/12 & \hspace*{1.5mm}672 & + & -- & + & -- & -- \\
  2 & \citet{kos95} & 1994/12 & 1007 & + & -- & + & -- & -- \\
  3 & \citet{sat98} & 1998/08 & 1264 & + & + & + & -- & -- \\
  4 & \citet{sat03} & 2001/12 & 3112 & + & + & + & + & -- \\
  5 & \citet{sat06} & 2001/12 & 3112 & + & + & + & + & + \\
 \hline
\end{tabular}}
\medskip
$^{a}$ lc -- light curves; i -- images.
\end{table}